Author's response to reviews

Title: A review of population-based prevalence studies of physical activity in adults in the Asia-Pacific region

Authors:

Rona Macniven (rona.macniven@sydney.edu.au)
Adrian Bauman (adrian.bauman@sydney.edu.au)
Marian Abouzeid (marian.abouzeid@greaterhealth.org)

Version: 3 Date: 15 December 2011

Author's response to reviews:

We have revised the paper, “A review of physical activity population-based prevalence studies in the Asia-Pacific region” following comments from the Associate Editor received in an email dated Wednesday 14th December 2011.

Comments: "I believe that the authors have adequately addressed the concerns raised by the reviewers. I wonder if we could ask them to reconsider the use of the word 'fugative' - it is now a well-known term and some simple editing could make the manuscript slightly more readable."

Author response: The word ‘fugative’ has been removed from the manuscript. The authors have made additional minor general edits to the manuscript to make it more readable, such as correcting one or two grammatical errors and reducing the length of some sentences.

We believe these responses provide sufficient answers to comments from the Associate Editor.